Received July 23, 2021; revised September 14, 2021; accepted October 26, 2021; date of publication November 12, 2021; date of current version January 5, 2022.

Digital Object Identifier 10.1109/TQE.2021.3127503

Teaching Quantum Computing to High-School-Aged Youth: A Hands-On Approach

PRASHANTI PRIYA ANGARA¹, ULRIKE STEGE¹, ANDREW MACLEAN¹, HAUSI A. MÜLLER¹, AND TOM MARKHAM²

University of Victoria, Victoria, BC V8P 5C2, Canada Ouantinuum Ltd., Minneapolis, MN 55422 USA

Corresponding author: Prashanti Priya Angara (pangara@uvic.ca).

This work was supported in part by the U.S. Department of Commerce under Grant BS123456, in part by the National Sciences and Engineering Research Council (NSERC) of Canada, IBM Canada Ltd., and in part by the IBM Center for Advanced Studies, and the University of Victoria.

ABSTRACT Quantum computing is an interdisciplinary field that lies at the intersection of mathematics, quantum physics, and computer science, and finds applications in areas including optimization, machine learning, and simulation of chemical, physical, and biological systems. It has the potential to help solve problems that so far have no satisfying method solving them, and to provide significant speedup to solutions when compared with their best classical approaches. In turn, quantum computing may allow us to solve problems for inputs that so far are deemed practically intractable. With the computational power of quantum computers and the proliferation of quantum development kits, quantum computing is anticipated to become mainstream, and the demand for a skilled workforce in quantum computing is expected to increase significantly. Therefore, quantum computing education is ramping up. This article describes our experiences in designing and delivering quantum computing workshops for youth (Grades 9-12). We introduce students to the world of quantum computing in innovative ways, such as newly designed unplugged activities for teaching basic quantum computing concepts. We also take a programmatic approach and introduce students to the IBM Quantum Experience using Qiskit and Jupyter notebooks. Our contributions are as follows. First, we present creative ways to teach quantum computing to youth with little or no experience in science, technology, engineering, and mathematics areas; second, we discuss diversity and highlight various pathways into quantum computing from quantum software to quantum hardware; and third, we discuss the design and delivery of online and in-person motivational, introductory, and advanced workshops for youth.

INDEX TERMS Computer science (CS) unplugged (CS Unplugged), education, entanglement, high-school-aged youth, measurement, qiskit, quantum computing, quantum computing games, quantum gates, quantum teleportation, qubit systems, superposition, teachers, training, workforce development.

I. INTRODUCTION

Many scientists and engineers consider quantum computing as one of the most interesting, exciting, and challenging topics in science and engineering, with enormous potential in numerous application areas [1]. In 2017, Marr wrote in *Forbes Magazine*: "Quantum computers will disrupt every industry. They will change [...] how we fight disease and invent new materials, and solve health and climate problems" [2]. With recent rapid advances in quantum technology, quantum computing has gained significant importance and has become one of the fastest-growing fields in the science, technology, engineering, and mathematics (STEM) areas [3]. Governments, funding agencies, and companies have been heavily investing in the development of quantum computing with the mission to realize hardware and software innovations that allow the full potential of quantum computing [4]–[9].

The accelerating progress in the area has now paved a way to educate a workforce for a diverse industry. Universities have begun to offer courses and degree programs at the graduate and undergraduate levels. Furthermore, quantum computing is now an active research area not just in physics but in disciplines such as chemistry, electrical and computer engineering, and computer science (CS). The quantum *computing* journey was initiated by Feynman in the early 1980s [10]. Since then, several important results in quantum algorithms have emerged [11]-[18], including Shor's factoring algorithm [19] and Grover's search algorithm [20]. Over the last 15 years, companies (e.g., [21]-[27]) have engineered the first real quantum computers and developed quantum development kits (QDK), such as IBM Qiskit [28], D-Wave Leap [29], Microsoft QDK [30], Google Cirq [31], Rigetti Forest [32], and Xanadu Strawberry Fields [33]. Today, we are in the noisy intermediate-scale quantum (NISQ) era [34] chasing quantum advantage with shallow circuits and hybrid quantum-classical algorithms. The need for continued research and development in quantum computing at all levels remains stark, and quantum skills are (still) rare to find [35]. The labor pool in quantum computing is anticipated to be interdisciplinary. From quantum software to quantum hardware, knowledge from physicists, engineers, computer scientists, software engineers, and mathematicians among others is essential for quantum computing [3]. Fortunately, there are many pathways into quantum computing from a variety of disciplines.

In addition to diverse skill sets, we require workforce diversity. It is predicted that "25% of the Fortune Global 500 will gain a competitive edge from quantum computing by 2023."¹ As such, the area of quantum computing presents a (likely short) window of opportunity to create a truly diverse pool of quantum computing talent. In this spirit, Asfaw, IBM's Global Education Lead, calls for quantum education programs to reach a diverse student body: "IBM realizes that quantum computing's success both in the laboratory and the enterprise relies heavily on our ability to expand educational opportunities to a broader diversity of students" [36]. According to Johnson [3], Brower-Thomas, Education Director, and Howard University Executive Director of the Center for Integrated Quantum Materials, posits: "So the question is are we preparing our K-12 to go to the schools that have a requisite curriculum that will prepare them to be in the [quantum] industry? I think, unfortunately, the answer is 'no'." Also, preparation in classical computing and computational thinking skills is still lacking for K-12, although education researchers have long advocated teaching (classical) computational thinking in all schools [37] and describe them as key skills not just for our 21st Century workforce but for all humans [38]. At the same time, the lack of gender diversity in the tech industry continues to be problematic [39], despite mounting evidence that more diverse teams are not just required for a healthy tech sector [40] but are also more successful financially. Research published by McKinsey in 2015 found that companies in the top 25% for racial and ethnic diversity are 35% more likely to have better financial returns than the corresponding national median, and companies in the top 25% concerning gender diversity are 15% more likely to have financial returns above their national

median in the corresponding industry [41]. In 2020, the United States National Center for Women and Information Technology (NCWIT) published statistics² on the proportion of women in computing in the student body in universities as well as the workforce in industry: While 57% of professional occupations in the U.S. workforce were held by women, only 26% held computing occupations, and women only held 18% of CIO positions in the Top 1000 Companies. This may not come as surprise when looking at the recipients of bachelor degrees: While 57% of bachelor degree recipients were women, only 21% women were recipients of bachelor degrees in computing in 2020, in 1985, 37% were recipients of bachelor degrees in computing. Results from a Microsoft study in the U.K. published in 2017 identified that the critical window to motivate girls for careers is between the ages of 11 and 15: they start gaining interest in STEM areas typically at the age of 12 or 13 and start losing interest shortly after they turn 15 [42]. One of the reasons for their choice to turn away from a tech career is apparently the lack of (perceived) equal opportunity in the workplace.

Since the field of quantum computing education is still in its early stages, it presents many opportunities to include diversity and gender balance in the field of quantum computing. We advocate that teaching quantum computing to youth should begin no later than in Grade 9 (i.e., students are about 14 years old). Through teaching quantum computing at this level, we hope to add information for career choice and strategies of entry into technology, specifically quantum computing. In addition to supporting career choice decision making, since quantum computing is quite different from classical computing, introducing a different way of thinking early might make the area of quantum computing more approachable. In his book Quantum Computing Since Democritus [43], Aaronson points out that quantum mechanics, the foundation of quantum computing "is a beautiful generalization of the laws of probability" and "while quantum mechanics was invented a century ago to solve technical problems in physics, today it can be fruitfully explained from an extremely different perspective: as part of the history of ideas in math, logic, computation, and philosophy, about the limits of knowable." We follow this ideology for our workshops and posit that a background in quantum mechanics or quantum physics is not required to understand fundamental concepts in quantum computing. We submit that understanding quantum computing can open doors to understanding quantum mechanical concepts [44]-[46]. For participation in our quantum computing workshops for high school students, we have no expectations regarding the students' prior knowledge-workshop attendance does not require a background in quantum computing, quantum physics, quantum mechanics, CS (including programming and algorithms), mathematical topics (such as linear algebra and complex

¹https://www.analyticsinsight.net/when-will-quantum-computingcome-to-mainstream/

²https://www.ncwit.org/sites/default/files/resources/ncwit_btn_07232020.pdf

numbers), or software development environments. Nevertheless, we do assume that the common K–8 syllabi have been covered by all the participants. This article presents our experiences in training high-school students with the goal to show different pathways into quantum computing as part of 1) the University of Victoria's HighTechU program [47] and 2) the IEEE International Conference on Quantum Computing and Engineering 2020 (QCE20) in collaboration with Honeywell Quantum Solutions. We will refer to QCE20 also as Quantum Week.

The rest of this article is organized as follows. Section II overviews relevant literature on quantum computing education and training with a focus on the delivery of quantum computing to youth, computer science unplugged (CS Unplugged) activities-hands-on activities teaching computing concepts without the use of a computer, and quantum computing games. We also took into account the K-12 curricula in Canada and the U.S., with a focus on British Columbia, Canada. Section III discusses the design of our workshops through activities-choosing from best practices-and lays out the developed activities, including activities for a potential new chapter of CS Unplugged [48]: Quantum Computing Science Unplugged. Section IV describes the goals and the overarching theme of our workshops. Section V introduces three workshop formats, and their respective design and delivery methods. Section VI discusses lessons learned from delivering workshops and student feedback. Finally, Section VII concludes this article and presents some of the materials used in delivering the workshops given in Section VII-A.

II. RELATED WORK

This section begins by summarizing ongoing work in quantum computing education efforts, including introductory textbooks and courses suitable for high-school students, CS Unplugged resources, and quantum computing games. Quantum computing education is developing fast around the globe. Notable developments include the White House Office of Science and Technology Policy and the National Science Foundation spearheading a partnership between the Federal government, industry, professional societies, and the education community called the Q2Work Program [49]. This program enables a range of training opportunities to increase the capabilities, diversity, and number of students who are ready to engage in the quantum workforce. It caters to K-12 education in quantum computing with outreach for middle and high-school students. We next discuss selected recent educational efforts in quantum computing primarily aimed at the K-12 demographic. The year-long course of the Qubit by Qubit initiative³ is offered with the goal to "increase diversity in the field of quantum" as well as "ensure that students [participating in the program] are prepared for the quantum workforce" and "to educate and inspire students to learn more about quantum computing." Qubit by Qubit is

offered collaboratively by IBM and MIT and makes quantum computing accessible to everyone, and specifically to high-school students. This course is an initiative by The Coding School⁴ and sponsored by IBM Quantum and Google Quantum AI [24], [50]. The book Dancing With Qubits [51] begins by providing a gentle introduction for those who are new to quantum computing, but soon delves deeper into the foundations and algorithms. It offers programming examples using Python and IBM Qiskit. The goal of this book is to make quantum computing approachable to people without a quantum physics background. University of British Columbia's K-12 Quantum Education Program⁵ offers educational resources and unique games (such as Qubit Jump! and Quantum Adventures! to learn quantum computing). In their article, Economou et al. [52] offer an educational outreach program for high-school students that provides handson knowledge of quantum circuits and algorithms and is based on the book Q is for Quantum by Terry Rudolph.⁶ This book allows students to learn some basic quantum computing concepts without the use of any linear algebra. The authors use marbles to represent qubits (with the color representing a state) and superposition represented by misty marbles. They also introduce Deutsch's algorithm through a game called Money or Tiger. Shoshany [53] presents an introduction to quantum theory for exceptional high-school students through a series of lectures that span four days at the Perimeter Institute at the University of Waterloo in Ontario, Canada. The lectures focus on building a mathematical foundation for quantum computing and ramp-up to advanced mathematical foundations related to infinite-dimensional Hilbert spaces. Mykhailova and Svore [54] provide an experience report teaching quantum computing as an undergraduate course through a practical software-driven approach. They rely on self-paced programming exercises called the Quantum Katas and observe that programming assignments help internalize theoretical concepts much better than relying on just lectures. The Qiskit Textbook embodies literate programming [55] and is an evolving online resource with complete Qiskit documentation [56]. It is an introduction to programming using Qiskit on the IBM Quantum Experience [57] while explaining quantum computing concepts and algorithms. Coding with Qiskit is a famous online series of videos introducing beginners to programming using Qiskit with Python and Jupyter Notebooks [58], [59]. Tappert et al. [60] overview their experience of teaching quantum computing as a graduate course as well as a one-week high-school workshop. They describe that for high-school students, the focus was on short lectures followed by hands-on programming utilizing the composer interface on the IBM Quantum Experience, to then run their programs on the quantum simulator as well as an actual quantum computer. Quantum Computing

³https://www.qubitbyqubit.org/

⁴https://www.codeconnects.org

⁵https://quantumcomputing.ubc.ca/education/k-12-education

⁶https://qisforquantum.org/

as a High School Module by Perry et al. [61] is a textbook with exercises for teaching quantum computing and serves as a bridge between popular articles and undergraduate texts. This work is intended as a one-week course for high-school students. Many exercises are given in the form of practice sheets that can be completed without computers. Billig's textbook on Quantum Computing for High School Students [62] introduces mathematical foundations of quantum computing, assuming high-school-level mathematics. They include selected linear algebra elements but intentionally leave out complex numbers. Billig argues that this simplification helps in conveying and visualizing quantum concepts. Quantum Computing the Easy Way [63] is a Udemy course that teaches quantum computing and introduces the basics of quantum computing and quantum algorithms. It is open to students from any background. The material is delivered using PowerPoint slides only and does not involve programming. Barabasi et al. [64] discuss student experiences using IBM's Quantum Experience platform in developing games based on quantum randomness. Salehi et al. [65] conduct an extensive study teaching introductory quantum computing using a CS oriented approach where quantum computing is viewed as a generalization of probablistic computing. They conduct two- or three-day workshops for students coming from diverse backgrounds. However, very few students were from a high-school demographic (only 19 students out of 317). CS Unplugged [48] emerged in the early 1990s⁷ and was first publicized via a 1998 manuscript entitled Computer Science Unplugged ... off-line activities and games for all ages by Bell et al. [66].8 Today, CS Unplugged is considered an educational method for introducing K-12 nonspecialists to computing concepts through hands-on activities that do not require the use of a computer [67]. The success of CS Unplugged activities has been demonstrated for learning computing concepts at the K-12 level. Nishida et al. [68] introduced CS Unplugged at the high-school level and concluded that students found the topics enjoyable and understandable. Taub et al. [69] evaluated CS Unplugged activities by conducting interviews with middle schoolers. While they found that not all set-out objectives were met by teaching CS concepts using only unplugged activities, they did confirm that students comprehend the computing concepts taught. Carruthers et al. [70], [71] describe the impact of unplugged instruction with Grade 6 students when teaching graph theory. Her findings indicate that students are capable of learning graph theory, and applying these skills when working on mathematical word problems. Rodriguez [72], [73] created assessment strategies that use in-class assignments and a final project and evaluated computational thinking principles with a well-defined rubric and notes from classroom observations. The results show that CS Unplugged activities can yield an increase in computational thinking.

A number of quantum computing games, based on quantum computing concepts, have been developed for educational purposes. Game-based learning is a learning strategy focused on achieving the particular objectives of given educational content through game play [74]. Games facilitate collaborative learning and positive interdependence and help us teach complex concepts engagingly [75]. The first quantum game played on an actual quantum computer, called Cat/Box/Scissors [76], was presented by Wootton in 2017 and is a quantum version of the popular hand game Rock/Paper/Scissors.9 Since then quantum enthusiasts have invented many different games around core quantum concepts, such as qubits, superposition, entanglement, interference, and measurement (e.g., Quantum Battleships [77] and Entanglion [78]). Several quantum games exploit quantum randomness. The state space of a quantum system increases exponentially with an increasing number of qubits thereby increasing the number of outcomes exponentially. For example, in Quantum Battleships, each qubit on a ship is a potential position, and bombs are implemented through 1qubit rotations. *qCraft* is a quantum take on Minecraft [79]. It is a single-player game where players discover materials and build structures in a procedurally generated 3-D world. Using qCraft, players learn superposition and entanglement by solving a series of puzzles. Quantum Chess is a reimagination of chess [79] and involves pieces that can be in a superposition of different squares on the board where measurements occur when a piece is attacked. Quantum Cats [80] is a single-player game developed by the Institute for Quantum Computing and teaches the differences between classical and quantum computing with the help of cats. Decodoku [81] is a puzzle available for example as iOS and Google Play app, which focuses on quantum benchmarking through quantum error correction: through the solving of puzzles, the quantum computer must be kept free of errors to keep running. Entanglion [78] is a two-person co-operative board game designed at IBM Research. It exposes players to the fundamental concepts of quantum computing such as superposition, entanglement, and gates. The goal is to build a quantum computer by visiting Galaxies to pick up components that can build a quantum computer.

III. EFFECTIVE TEACHING METHODS

This section discusses selected effective methods that we use to teach quantum computing to high-school students. The British Columbia Teaching Federation (BCTF) [82] outlines effective teaching strategies and classroom complexity for high-school students and describes the principles of learning as follows.

- 1) Learning requires the active participation of the student.
- 2) Students learn in a variety of ways and at different rates.

9https://en.wikipedia.org/wiki/Rock_paper_scissors

⁸For a revised edition see classic.csunplugged.org.

 TABLE 1 Effective Teaching Strategies Mapped to Activities

	Experiential	Independent	Indirect	Interactive
	Learning	Study	Instruction	Instruction
Quantum doughnuts	\checkmark			\checkmark
Find Your Quantum Partner	\checkmark			\checkmark
Bloch Sphere	\checkmark			
Measurement/Probability	\checkmark			\checkmark
Practice Sheets		\checkmark	\checkmark	
Entanglion/Games	\checkmark			\checkmark
Programming			\checkmark	\checkmark

TABLE 2 Quantum Computing Concepts Mapped to Activities

	Quantum State	2-qubit System	Gates	Superposition	Measurement
Quantum doughnuts	√	\checkmark		✓	\checkmark
Find your Quantum Partner	\checkmark				
Bloch Sphere	\checkmark		 ✓ 	\checkmark	√
Measurement/Probability	\checkmark			\checkmark	√
Practice Sheets	\checkmark	\checkmark	 ✓ 	\checkmark	√
Entanglion/Games	\checkmark	\checkmark	 ✓ 	\checkmark	\checkmark
Programming	\checkmark	\checkmark	 ✓ 	 ✓ 	\checkmark

3) Learning is both an individual and a group process.

BCTF's teaching strategies include direct instruction, experiential learning, independent study, indirect instruction, and interactive instruction. The design of our quantum computing workshop incorporates a subset of these strategies as participant activities. Our activities are categorized into unplugged activities, practice sheets, quantum games, and programming. The following sections explain the particular activities in detail. Table 1 describes a mapping of BCTF's teaching strategies to each of the activities, while Table 2 presents a mapping of the quantum computing concepts covered in the corresponding activities.

A. UNPLUGGED ACTIVITIES

We describe our unplugged activities designed to support the teaching of quantum computing concepts.

Qubit doughnuts: A qubit or a quantum bit is the basic unit of quantum information. We use images of a doughnut and a doughnut-shaped pillow to demonstrate a qubit.¹⁰ Qubits can be in state |0⟩ or |1⟩, or a linear combination of both. The sprinkled side of the doughnut represents |0⟩ and the plain side represents |1⟩. Superposition is shown to be a spinning qubit that stops at the sprinkled side or the plain side upon measurement.

We also use doughnuts to demonstrate the *concept of entanglement:* Two assistants hold up and spin the two doughnuts at opposite ends of the classroom (i.e., entangled qubits can be separated by a large distance). When measuring the state of one of the doughnuts, the two assistants stop the spinning and display opposite sides of the doughnuts (i.e., plain and sprinkled) facing



FIGURE 1. Find your quantum partner activity.

the class—simultaneously. While no classical analogy is perfect to explain quantum mechanical phenomena, we believe that qubits represented as doughnuts provided a memorable visualization for the students. Participants are asked to reconstruct a quantum state given a set of measurement outcomes. In other words, the students have to decide whether the doughnut's displayed side (i.e., the qubit state) is measured as plain or sprinkled (i.e., $|0\rangle$ or $|1\rangle$).

- 2) Find your quantum partner: The "find your quantum partner" activity demonstrates quantum states and their properties. The activity involves unicolored balls of two different colors with numbers written on them (cf. Fig. 1). The numbers correspond to the amplitudes of quantum states in a qubit. Each student is given one ball—they have to find a person with a ball that is of a different color, and marked with an amplitude such that the sum of the squares of both amplitudes equals 1. This activity also doubles up as an ice-breaker activity for the participants as there is an element of interaction involved.
- 3) Bloch sphere—Drawing on a balloon: The Bloch sphere is a geometrical representation of the pure state space of a qubit. For this unplugged activity, we show the Bloch sphere as a balloon, demonstrating different states a qubit can be in. The north pole is labeled $|0\rangle$

¹⁰The idea of representing qubits using doughnuts was inspired in a quantum computing talk by Jessica Pointing. See http://www.youtube. com/watch?v=YIDdPbnnqsA

and the south pole is labeled $|1\rangle$. If, for example, a Pauli-X gate is applied, using a marker a line is drawn from the north pole to the south pole indicating a state change from $|0\rangle$ to $|1\rangle$. Alternatively, participants are given balloons and are instructed to mark different states. Concepts related to states, gates, and rotations can be demonstrated using this activity.

- 4) Measurement and probability: The outcome of measuring a quantum system is probabilistic. In British Columbia, probability theory is absent from the K–11 mathematics curriculum; concepts are introduced in a statistics course in Grade 12.¹¹ Thus, participating students typically only have an intuitive idea about the notion of probability. In our youth workshops, we illustrate the intuition behind probability, uncertainty, and randomness with the use of a black box (a black colored box) containing unicolored balls of two different colors. The activity is described as follows.
 - 1) Setup: Box with seven $|0\rangle$ and three $|1\rangle$ balls.
 - 2) Goal: Determine the more likely state of a randomly picked ball from the box.
 - Process: Blindfolded (or without looking into the box) pick a ball and write down whether it is a |0> or a |1> ball. Place the ball back into the black box. Repeat.
 - 4) After enough draws, about 70% of the time |0⟩ is obtained, and 0 is observed as the more likely state. In addition, we vary the experiment by switching up the ratio of black and white balls in the black box.

B. PRACTICE SHEETS

Between sessions, participants are given practice sheets to reinforce the learning of concepts. Concepts included for practice sheet activities are quantum states, entanglement, 2-qubit systems, and quantum gates. The quantum states' practice sheet includes a list of valid and invalid quantum states represented using Dirac notation.

Students have to work out which states are valid and which ones are invalid. This exercise extends to two-qubit systems, where students are presented with four amplitudes instead of two. Not only do the students learn the concept of normalization through these practice sheets but also get comfortable with the Dirac notation. Practice sheets for gates involve calculating the different outcomes when one- and two-qubit gates are applied to specific qubit states. This sheet also includes a section on measurement where students are asked to write down the most likely outcome when measuring specific quantum states.

C. PROGRAMMING

While the unplugged approach takes students away from computers, programming plugs them right back in. Concepts

¹¹http://curriculum.gov.bc.ca/curriculum/mathematics

in quantum computing are introduced using Jupyter Notebooks and Qiskit. We chose the IBM Quantum Experience platform for the following reasons: First, IBM provides quantum computers and simulators free to use. Second, the circuit composer is a user-friendly graphical tool that allows us to design and run quantum circuits without any prior programming knowledge. Third, Qiskit and Jupyter notebooks are integrated into their web-based cloud platform, making it easy to use and set up. Fourth, students can continue working at home using their IBM Quantum Experience account setup during the course and demonstrate their quantum computer programs to family members.

Students are given skeleton Jupyter notebooks to import into the IBM Quantum Experience. A handy cheat sheet is provided, comprising a list of some basic quantum operations such as creating, drawing, and measuring a circuit, and performing gate operations. The first notebook introduces the students to Jupyter notebooks. Here, they learn how to differentiate between code cells and markdown cells, create cells, write markdown, and execute cells. In the initial notebook, we ask students to write down their expectations for the workshop and read it out to other participants. In addition to familiarizing the students with the Jupyter notebook environment, these activities also remove students' initial hesitation to interact with other participants and help to make the workshop more interactive and fun for students and instructors alike.

D. QUANTUM GAMES

Entanglion is a two-person co-operative open-source quantum board game developed by IBM [78] that introduces, or enforces, fundamental concepts of quantum computing such as quantum states, gates, and entanglement. Its goal is to build a hypothetical quantum (space) ship by going into entangled planets (states) and collecting features to build the ship. For ships to land on an entangled planet, it is essential to put them in superposition. This is achieved using gate cards. We normally schedule Entanglion at the end of the workshop, when it becomes too overwhelming to learn more new topics. This allows students to think about quantum computing concepts in a different, fun, and engaging way.

E. ONLINE TOOLS

The delivery of online workshops involves additional tools that are useful for fostering collaboration and interaction.

 Online lectures via Zoom:¹² Every session is hosted as a Zoom meeting so that participants can access the workshop virtually. To promote interactivity and collaboration, students are divided into groups of four or five students. Zoom's feature of breakout rooms is used for team and group activities. A team of moderators cycle through the breakout rooms to answer any questions that the students might have. Zoom's feature of

¹²https://zoom.us/

polling allows the organizers to ask multiple-choice questions to attendees. Therefore, it is used intermittently in lectures to make sure that students follow the material. The anonymity of polls also helps students be more comfortable answering the questions.

2) *Quirklogic Workbooks:*¹³ Quirklogic provides remote education learning solutions where digital handwriting is used on their e-ink devices (such as Papyr and Quilla) and can be accessed using an application on iOS/Android platforms¹⁴ as well as a webbased application.¹⁵ Documents are stored as Quirklogic workbook files that can be downloaded as PDFs. Files can be uploaded as PDFs, which get translated into Quirklogic workbook files. Since the notation and mathematics in quantum computing cannot be typed easily (such as the Dirac notation or matrices), digital tools that use handwriting are helpful. If available to them, students are recommended to use the application on iOS/Android devices (e.g., the iPad) because of their touch capabilities and the fact that many students or their parents own such a device. The first Quirklogic notebook is used as an icebreaker activity where students can write in their name and their expectations of the workshop. This allows them to get familiar with the platform and encourages collaboration. The remaining notebooks contain practice sheets (which in an in-person setting are printed and handed out to students). For the breakout-room activities, each group of students is assigned to one shared Quirklogic workbook on which they can collaborate and figure out the solutions to the questions asked.

This section introduced the effective teaching methods, activities, and tools used in in-person and online settings. The following section describes pathways into quantum computing by discussing some quantum computing challenges and recommended classical skills to delve into quantum computing.

IV. PATHWAYS INTO QUANTUM COMPUTING

This section introduces the overarching theme used in our workshops (workshop design and delivery are described in detail in Section V). We introduce quantum computing to a high-school-aged audience with the following goals in mind.

- 1) Students learn hands-on how quantum computing works.
- 2) Gain understanding about the larger context of quantum computing and the breadth of fields that are crucial for the future of quantum computing.

Students are provided with an overview of the workflow of a quantum computing program to highlight different parts of

¹⁵[Online]. Available: https://portal.quirklogic.com/



FIGURE 2. Software stack linking users to quantum hardware.

the quantum software stack linking users to quantum hardware. In the following, we provide an overview of quantum computing systems with an emphasis on hardware. The goal is to introduce some of the science and technologies required on the other side of the programming interface. Introducing students to the breadth of engineering disciplines necessary to build a quantum computer is intended to stimulate their interest in STEM-related classes. Areas that provide a pathway to quantum education include electrical and computer engineering, CS, chemistry, and physics. A summary of the quantum stack consisting of the user interface, Application Programming Interface, Compiler and operating system, and Quantum Hardware is given below.

Quantum computers, such as their classical counterparts, are built in layers. The software stack model depicted in Fig. 2 introduces a high-level generic model used by various quantum systems today.

User: The user writes the quantum program in a high-level language such as the Open Quantum Assembly Language (openQASM) standard. Alternatively, users can also take advantage of libraries (such as Qiskit, Microsoft QDK, Rigetti Forest SDK, or Pennylane) to convert their programs to the openQASM standard. This allows the user to describe the particular quantum circuits constituting the quantum algorithm. OpenQASM also supports linking quantum and classical computations in a hybrid quantum-classical model. In our workshops, students are introduced to the IBM Quantum Experience where they can implement quantum programs using the circuit composer or a Python program using the Qiskit library (as a Jupyter notebook). The IBM Quantum Experience supports Jupyter notebooks natively, that is, students can access these from the Quantum Experience cloud platform in a web browser.

¹³https://www.quirklogic.com/

¹⁴iOS/Android application: https://www.quirklogic.com/products/quirk-logic-connect



Hadamard Gate & Measure

FIGURE 3. Processing of a quantum circuit as it moves through the quantum stack.

Application programming interface: Quantum computers are large, expensive, and complex hardware systems. Thus, users require a method to submit the programs created in openQASM to the quantum computer. The application programming interface (API), often implemented as an internetaccessible resource, provides the path. The API handles account creation, user login, job submission, job status, and delivery of the job results. In the workshops, students submit jobs to a quantum simulator or IBM quantum computer using the IBM quantum experience.

Compiler and operating system: The quantum computer cannot directly run the openQASM circuit. First, the openQASM code must be translated or compiled into machine code which the quantum hardware can execute. The "operating system" function schedules resources, performs calibration when necessary, controls running the quantum job then validates the results and sends the result back to the user via the API.

Quantum Hardware: The hardware layer consists of the quantum processor (e.g., ion trap manipulating ytterbium ions) and the controls that turn ON/OFF lasers, turn ON/OFF microwave signals, control magnetic fields, regulate radio frequency (RF) energy and maintain temperatures and vacuum around the quantum gates. Fig. 3 provides an example of how a quantum circuit is processed as it moves through the layers shown in Fig. 2. Within the world of classical computing, the hardware technology is mature and most classical processors as those used in laptops or cell phones use a chip built using complementary metal–oxide–semiconductor

field effect transistors, CMOS for short. In contrast, quantum computing technology is new, and with several competing technologies. Experienced engineers and computer scientists often find the probabilistic model of quantum computing requires new approaches. We introduce two approaches to quantum computing hardware to provide students with an appreciation for science and engineering, which underlies the machine that runs their openQASM circuit. The science of quantum computing involves quantum mechanics and an understanding of electron behavior. Engineering in the context of quantum computers is about taking science and making it reproducible, reliable and costeffective.

Superconducting qubits: D-Wave Systems are a Canadian company that produces quantum computers focusing on the quantum annealing process. D-Wave uses the Superconducting QUantum Interference Device (SQUID) as the building block. The magnetic spin states of SQUID can be placed into quantum superposition. These building blocks are then connected via "couplers" to represent the circuits submitted by users.

IBM uses superconducting qubits built upon Josephson junctions. Both D-Wave and IBM perform quantum computations then use classical electronics to read out the state of the qubits in order to return 1 s and 0 s to the user.

Trapped ions: Honeywell Quantum Solutions, IonQ, the Universities of Sussex and Waterloo work on building qubits by trapping a small number of ions and then manipulating those ions using lasers, magnetic fields and microwave energy. Quantum charge-coupled device traps hold ions within an RF field and use potential wells generated by electrodes to move the ions to different locations on the trap. Trapped ions have longer coherence times and lower error rates than superconducting qubits.

The challenges for both hardware approaches include scaling the number of qubits, reducing the errors introduced into the system, and providing longer coherence times. Some specific challenges and courses students may take to prepare themselves to support research overcoming the challenges are listed in Table 3. The table also depicts which part of the quantum stack the challenges belong to.

V. WORKSHOP FORMATS

Our workshops are classified into three categories based on the content delivered: motivational, introductory, and advanced. They can be delivered in-person or online. Topics are grouped together as modules, and a collection of modules forms a workshop.¹⁶

A. MOTIVATIONAL WORKSHOPS

The motivational workshops primarily aim to generate interest and provide pathways to a career in quantum

¹⁶All materials related to our workshops can be accessed on GitHub: https://github.com/pangara/TQE-HighTechQ.

TABLE 3 Engineering Challenges and Recommended Education

Quantum stack	Quantum computing challenge	Recommended classical skills
User and API Interface	Developing quantum algorithms and interpreting their results	Linear algebra, probability theory and classical algorithms
	Creation and execution of hybrid programs combining classical and quantum computing	Computer Science, programming, software engineering, algorithms
Compiler and Operating System	Compiling complex quantum circuits to execute efficiently on quantum processor	Computer science: Algorithm development, compilers, sorting and routing
	Scaling up digital control systems to handle 10,000 to a million qubits	Digital system design, control systems and computer architecture
Quantum Hardware	Providing extremely low noise analog designs for control and measurement Design of scalable, stable laser systems and optics	Analog design, calculus and radio frequency (RF) design Optics, photonics, materials science and physics
	Designing components and operating them at cryogenic temperatures near absolute zero	Physics, mechanical engineering and materials science

computing. Such a workshop can be used as part of other, longer, workshops—described in the following—as a motivation module. Motivational workshops are designed to be delivered in less than two hours.

We begin such a workshop interactively by asking the audience what they know about quantum computing. Then the field of quantum computing is introduced by relating it to quantum physics and CS. This is followed by describing computational problems that current classical computers cannot solve, but potentially could be solved by quantum computers. This provides a segue into careers and pathways to get started in quantum computing. We then introduce superposition, entanglement and measurement using qubit doughnuts (cf. Section III). Finally, we demonstrate running quantum circuits on the IBM Quantum platform using the circuit composer and available quantum computers to show the accessibility of quantum computers today.

B. INTRODUCTORY WORKSHOPS

This section describes our experience in delivering an introductory workshop on quantum computing, delivered in 6–8 h. All course materials are carefully tailored to the high-school level. This workshop introduces basic principles of quantum computing by providing a hands-on experience on quantum computers and simulators, using Jupyter Notebooks and Qiskit. A secondary aim of this workshop is to feature our unplugged quantum computing activities, thereby contributing to CS Unplugged in general and a "Quantum Computing Unplugged" chapter in particular. A brief rundown of our introductory workshop is depicted in Table 4 and described in detail in the following paragraphs.

1) MOTIVATION MODULE

This module provides a brief introductory presentation, motivating participants regarding quantum computing. This is similar to our motivational workshop, but we leave the basic TABLE 4 Sample of an Introductory In-Person Workshop Schedule (8 H)

09:00	Introduction and Motivation
09:30	Account Setup on IBM Quantum Experience
10:00	Break
1015	Introduction to Jupyter Notebooks, Python, and Qiskit
11:00	Qubits, Gates, Superposition, and Measurement
12:30	Lunch
13:00	Hands-on Programming
14:30	multi-qubit Systems and Entanglement
15:30	Break
15:45	Entanglion
16:45	Closing Remarks, Feedback, and Outlook

concepts such as superposition and entanglement for later. We assume that most participants would have only heard of the term "quantum computing" but not know anything beyond the fact that it is a cool technology. This presentation highlights some promising applications (such as molecule simulation) and helps students understand why some problems are classically intractable. The main goal is to educate students regarding the existence of quantum computing technology and to demonstrate how readily accessible the technology is to everyone.

2) QUANTUM BASICS MODULE

This crucial workshop part introduces participants to fundamental concepts of quantum computing. This session alternates between direct instruction and unplugged activities, to foster participant engagement. This part integrates the qubit doughnuts and the unplugged activities "Find Your Quantum Partner" and "Measurement and Probability." Moreover, participants are given practice sheets to reinforce the understanding of the notions of quantum state and measurement.

3) BASIC PROGRAMMING MODULE

Participants are introduced to the IBM Quantum Experience [57] early on so that they can start experimenting with

TABLE 5	Schedule of	fa	Virtual	Workshop	(Quantum	Week)
---------	-------------	----	---------	----------	----------	-------

Session	Module		
	What is Quantum Computing? A quick introduction		
Motivation	Main concepts: Qubits, Superposition, Entanglement, Measurement, Quantum Gates		
	Hands On: Circuit Composer		
Different kinds of quantum computers	Ion-Trap Quantum Computing		
	Dive into D-Wave		
	Setup for Session 3		
	Multi-qubit systems		
Programming Quantum Computers	IBM-Quantum Experience and Qiskit		
	Teleportation		

quantum programs. We help students create accounts to access both Quantum Composer and Quantum Lab on the IBM Quantum Experience. Participants are then introduced to the basics of Jupyter Notebooks including markdown and code cells where they can enter some entry-level Qiskit quantum code, such as creating and drawing a quantum circuit.

4) INTERMEDIATE QUANTUM COMPUTING MODULE

After the hands-on programming session, participants are introduced to 2-qubit systems, and in general multiqubit systems. Students are given practice sheets to understand the concepts of 2-qubit systems and entanglement. Similar to the practice sheets for 1-qubit systems, students have to check the validity of a 2-qubit quantum state. They are asked to predict the outcomes of measurements for different quantum states. Finally, students are given the practice sheet to create different kinds of entangled states. The qubit doughnuts are used as a prop to explain these concepts.

5) HANDS-ON PROGRAMMING MODULE

In this module, participants are provided with skeleton Jupyter Notebooks containing boilerplate code that they are asked to fill in, based on the topic discussed. With the help of a handy cheat sheet, the notebooks are easy to follow. This module builds upon the quantum basics module, and topics such as Entanglement, Teleportation, and Superdense coding are programmed by participants.

6) GAMING MODULE

Workshops can be concluded by playing a quantum game that reinforces some of the concepts learned earlier in the workshop. An example is the board game Entanglion. Students are paired up and provided each with a copy of the board game *Entanglion*. While playing *Entanglion* does not require the players to have a background in quantum computing, the game provides additional learning reinforcements for the basic quantum computing concepts acquired earlier in the day.

Our Online workshop: This part is based on our experiences delivering an online introductory workshop titled "From Qubits to Quantum Teleportation" held at the IEEE International Conference on Quantum Computing on October 12, 2020. Participating students were Grade 8–12 students in Victoria, BC, Canada, and Broomfield, Colorado.

The main aim of this particular workshop was to stimulate the interest and awareness of high-school-aged youth and introduce high-school teachers to quantum computing as a possible field to teach. Since there are limited materials available that are targeted toward high-schoolers in the realm of practical quantum computing [83], this workshop aims to demonstrate how quantum computing can be a part of teachable component in high school.

We deliver the workshop in three modules, with each module lasting 1.5 h. An outline of the topics taught through the day is given in Table 5. Since this is an online workshop, it is easier to involve guest speakers who might not be from the same city where the workshop is hosted, which helps in diversifying the kind of content that can be delivered.

As usual, the first module introduces students to quantum computing. In this specific workshop, we started the session with a talk by a high-school student from The Knowledge Society¹⁷ (an organization that exposes youth to emerging STEM fields) with the aim to inspire high-school students and show them that (1) it is indeed possible to learn quantum computing in high-school (2) it is an exciting topic that they might find interesting to study further at university.

The content in the first module is similar to the content delivered in the motivation and quantum basics module of the in-person introductory workshop (i.e., students are introduced to single-qubit systems, superposition, measurement, and gates). In the second module, students learn about the different kinds of quantum computers that are available today with talks on superconductivity and ion-trap quantum computer technologies. In the third module, students encounter multiqubit systems, entanglement, and programming with Qiskit using the circuit composer (mapped to the Intermediate Quantum Computing Module).

C. ADVANCED WORKSHOPS

An advanced workshop allows participants to gain a deeper understanding of the basic quantum computing concepts than an introductory workshop allows. In the introductory workshop, the basic quantum computing concepts include qubits, one-, and two-qubit quantum systems, superposition, gates, entanglement, quantum arithmetic rules, and the bra–ket notation. The advanced workshop is designed to develop the mathematical foundations of quantum computing as well as to verify the correctness of quantum arithmetic by means

¹⁷https://tks.world/

Part 1	Part 2
Introduction and Motivation	Review of Part 1
Account setup on IBM Quantum Experience	1-qubit systems/entanglement
Introduction to Jupyter Notebooks, Python and Qiskit	Math: multi-qubit systems and entanglement
Linear algebra, vectors, matrices, complex numbers	Qiskit: programming/Jupyter with multi-qubit systems
Qubits, gates, measurements (math)	Introduction to quantum algorithms
Qubits, gates, measurements (programming)	Superdense Coding/Teleportation/Deutsch-Josza
Entanglion	Implementation
Closing and Feedback	Closing and Feedback

TABLE 6 Advanced Workshop Schedule (Sample)

of vectors and matrices. Additionally, the workshop introduces students to some notable quantum algorithms that demonstrate quantum advantage. Delivery of an advanced workshop takes about 16 h and can be delivered in two or more days.

In the first part of the advanced workshop, students are introduced to basic linear algebra, including complex numbers. As discussed earlier (cf. Section II) in his book, Billig introduces mathematical foundations of quantum computing without touching on complex numbers [62]. Similarly, we designed our advanced workshops to only mention the concept of complex numbers but avoid delving deeply into them—as most basic properties can be explained using real numbers only. Mathematical foundations are primarily required for describing the bra–ket notation for qubits and gate operations. We designed practice sheets for the same.

- 1) Understanding vectors and matrices: unitary matrices, transpose, addition, multiplication (Part 1).
- Converting vectors to and from bra-ket notation (Parts 1 and 2).
- 3) Tensor products (Day 2).
- 4) Mathematical operations when applying gates to qubits (Parts 1 and 2).

Items above with both Parts 1 and 2, signify that practice sheets are planned for both single and multiqubit systems. Tensor products come into play for multiqubit systems, covered in Part 2. In the second half of Part 2, we work through the details of significant algorithms such as Superdense Coding, Quantum Teleportation, or the Deutsch-Josza algorithm. These are taught using Jupyter Notebooks and Qiskit, where a skeleton of the program is given to them to complete and run. In their textbook, Perry *et al.* [61] outline teaching quantum computing to high-school students over one week. The textbook involves several exercises, some of which we think are useful additions when delivering our advanced workshop (e.g., linear algebra, entanglement, gate operations, or teleportation). Table 6 provides an outline of our advanced workshop.

VI. DISCUSSION/LESSONS LEARNED

This section describes the observations, feedback, and lessons learned from our introductory online workshop (held in 2020 at Quantum Week in collaboration with Honeywell Quantum Solutions) and our introductory in-person workshop (delivered by HighTechU at the University of Victoria).

A. FEEDBACK

Six students completed the Quantum Week Workshop Survey, with an average respondent age of 16. While 80% of respondents had no prior knowledge about quantum computing prior to the workshop, all had studied either linear algebra or complex numbers before entering the workshop. None had studied both. Using doughnuts for teaching the concept of quantum entanglement was reported as the most effective activity to aid in student understanding of quantum computing concepts, with 80% of respondents indicating that the activity had been helpful in their understanding of concepts. As a result of the workshop, students rated their understanding highest in how gates operate (Hadamard, Pauli-X, and CNOT gates) and lowest in entanglement and the teleportation algorithm discussed at the end of the workshop, with similarly low levels of comfort reported with the tools introduced in the workshop such as Qiskit and Jupyter Notebooks. Overall, respondents rated the workshop with a high overall average score of 4.2 out of 5, with 60% of respondents indicating they were likely to recommend the workshop to a friend. These results correlated with those from the introductory in-person quantum computing workshop delivered by HighTechU in January 2020 at the University of Victoria, where an average score of 4.2 out of 5 and a 80% recommendation rate was recorded for a group of 18 respondents and where students reported that "the next revolution in the world of computer systems will require yet another set of completely new skills" and that "Quantum computing is at the forefront, and my generation will be able to achieve major breakthroughs."

Online versus offline teaching: The online teaching environment did not allow for instructors to see the faces of some of the students who decided not to turn on their cameras for most of the time. For privacy reasons, we made it optional for students to turn on their cameras. Without the ability to read the students' body language, it was difficult to determine how well students were grasping the materials. This suggests that the structure should be modified to stimulate more interaction with the students and obtain more feedback. For example, rather than a long lecture, the materials could be broken into short segments. Following each segment, the instructor could ask the students a few questions to gauge their understanding. Simply asking the students if anyone has questions may not produce adequate feedback if students are not comfortable speaking up.

B. LESSONS LEARNED

The quantum computing workshops discussed in this article grew out of HighTechU [47], an innovative learning community and skills incubator for high-school-aged youths at the University of Victoria in Canada. In addition to the highschool workshops, we delivered half-day events for middleschoolers as well as the general public during UVic's 2020 IdeaFest.¹⁸ While this article mainly reports on the lessons learned from the high-school events, the other events contributed to our understanding of how to engage people in quantum computing.

While the background and age of workshop participants varied, by and large, the students were highly motivated and eager to engage in quantum computing. The list below includes specific lessons learned during the design, development, and delivery of our quantum engagements with youths.

Teaching strategies: As described earlier, we incorporated a mix of teaching strategies, including direct instruction, group activities, individual practice activities, and games. Mixing up these strategies helped considerably in keeping the students engaged and creating a lively day.

Engagement and active participation: Activities that foster interaction among the participants were critical for student satisfaction and overall experience. The first few activities —*Hello Jupyter*, where students could express their Workshop Expectations and *Find Your Quantum Partner*—served as effective icebreakers and nicely stimulated interaction.

Age diversity: While this workshop was intended for highschool-aged youths, there was considerable interest by other age groups—older (e.g., graduate students) and younger siblings signed up. Initially, we thought that this diversity of age groups might be an impediment but it actually generated synergy and fostered collaboration.

Props and handouts: Students clearly enjoyed props, especially the qubit doughnuts we used to explain entanglement. In their feedback, students mentioned that the usage of doughnuts helped them understand the concept of entanglement better. Reinforcement learning through practice sheets was quite popular and effective. Most of the students took these printed sheets home with the intention to engage their family.

Jupyter notebooks: Most students had little or no programming experience. Jupyter Notebooks are effective to communicate concepts in a step-by-step manner. Separating the introduction of "text"—and "programming code"-cells was critical to get started. Providing code fragments, templates, and complete solutions is important for a seamless and satisfying experience.

Quantum programming on the IBM Quantum Experience: The IBM Quantum Experience is an excellent platform for

the onboarding of students. The Qiskit software development kit with its Composer and Python programming interfaces is easy to learn. Our "Gates glossary" and "Qiskit cheat sheet" answered many questions along the way.

Reinforcement learning through games: Entanglion with its instructions and rules is an interesting game but challenging to learn after a long day when the students have already reached information overload. While the participants enjoyed playing *Entanglion*, they had difficulties identifying the quantum concepts they learned during the day in the *Entanglion* game. We think that this may be because there are many rules for the game and students had less time to play it. Given more time, they might appreciate it more.

VII. CONCLUSION

The evolution of both the quantum computer and the area of quantum computing is often compared with the evolution of the classical computer-more than 200 years passed since the invention of the first (mechanically programmable) computer by Babbage in 1800, and Alan Turing's conceptual model of the Turing machine, as well as the arrival of the first digital computer in the 1930s, to today's readily available access to computation power, via devices, such as smartphones and access to powerful computing machines via cloud access. The development of quantum computers and their uses is-in comparison-much faster, also since quantum computing research benefits on many levels from the developments in classical computing. The quantum computing journey itself started in 1981. In 1998, the first physical model of a quantum computer was presented [84]. As described, over the last 15 years, companies did not just engineer a variety of quantum computers but also designed QDK. These as well as cloud access to quantum computers and simulators is what accelerates quantum computing today. IBM's recently published quantum computing roadmap envisions moving speedily from the current era of NISO devices to the era of universal fault-tolerant quantum computers by 2023 [85]. Next to qubit-technology development, the focus will be on seamless integration of classical and quantum computing devices, as well as the evolving development of the quantum-classical software stack. This will ensure that quantum computing is seamlessly accessible as a service for application-level users. For this vision to come to fruition, a diverse workforce in quantum computing is needed.

This article discussed our design of different kinds of quantum computing workshops for high-school students. Today, different pathways into quantum computing are possible. Exposing high-school students to quantum computing concepts and possibilities early will enrich their decisionmaking when choosing a career path, which begins with selecting courses in the years before finishing high school. The implications for early exposure to quantum computing will likely positively impact the diversity of the large workforce needed in this area. As laid out above, diversity is needed in all dimensions, including background and specialization, as well as gender diversity.

¹⁸https://www.uvic.ca/ideafest2020/

All areas that interplay with the development of quantum computing, be it on the software or hardware side and the engineering or science side, have in common that these areas are under-represented by women. The rapid advancements in and the promise of quantum computing and its impactful applications are an opportunity to build a truly diverse quantum computing research and development workforce. Thus, the area in its current development stage presents an unprecedented opportunity to advertise science and technology areas to a diverse population. We believe that, since the diverse education of quantum computing experts allows a rather seamless transition into many other areas in tech, a more diverse quantum computing workforce can also positively impact the diversity of the tech workforce in general. We submit that, to reach women, quantum computing should be introduced early, ideally no later than Grade 9, the earliest high-school grade.

Our own intense efforts educating youth in quantum computing in the form of workshops with high-school-aged youth (grades 9-12) are short-term in nature. At the same time, our participants in our HighTechU community have the opportunity to engage in the longer term with a broad variety of themes in technology as well as mentors from the industry. We conducted preliminary formal evaluations for both the online and in-person workshops with 6 and 18 participants, respectively. Since these workshops were conducted over the course of a day, feedback was gathered from the students in the form of an exit survey where they were asked to evaluate themselves by rating their understanding of the different topics discussed in the workshop. To gain a more thorough formal evaluation, an updated survey should include short answer and multiple choice questions to test the students' understanding. This research is a part of an ongoing study where we continually improve our material based on the feedback of students and our own observations. The size of a cohort has been limited to 30 students so that there can be more effective attention given to each student. While the limited number of participants is not representative of the demographic, it provides a good starting point. One way to move forward is to conduct multiple iterations of the workshops and, subsequently, present this material as a course at the high-school level, which will increase the size of our case study. Given more time in the workshops, evaluations can be conducted for as group activities in the same cohort which probe their understanding of the material. Surveys can also be conducted over a period of time to test the students knowledge retention.

While there exist other longer-term efforts engaging youth in quantum computing in Canada (see Section II), we believe that an even broader effort directed through schools should not only be encouraged for a wider reach but is also feasible and practical.

Successful efforts in teaching quantum computing to youth around the world, such as our workshops, demonstrate that they can help inform young students when having to choose electives in high school as they can show a path for a future career. Early exposure to career possibilities and paths toward that need are critical to be presented sufficiently early. Both classical and quantum computing concepts and computational thinking skills are a way to train and deepen problem-solving skills that are beneficial for everyone.

So far, exposure to any form of quantum computing in school can be limited to chemistry or physics courses that some students in high school choose to take, and which content can strongly depend on schools and teachers. As discussed, despite the recognized need and benefits for all humans [38], classical computing and computational thinking skills are still lacking in our schools, as access to such materials is still not guaranteed for every student in K–12.

So far, to the best of authors' knowledge, the Computer Science Teachers Association (CSTA), the preeminent American Computer Science education association behind initiatives like the K–12 CS Standards,¹⁹ has demonstrated limited engagement in the area of Quantum Computing Education preuniversity, despite launching the AI4K12 working group in 2018 to explore how a similarly leadingedge technology—Artificial Intelligence—could be introduced into the elementary and secondary curriculum [86].

We advocate to refresh and widen the efforts of the integration of computational thinking skills into our K–12 curricula and include basic concepts from quantum computing in addition to concepts from classical computing. Quantum computing concepts are not only an asset to the problem-solving skill set of each individual, quantum computing can also serve as motivation to teach other foundational skills, including when working with probabilities, the manipulation of matrices, and complex numbers.

Exposing students to quantum computing through project work is another possibility. Accessible and always improving online resources, and free access to quantum computers support such endeavors just as much as the possibilities enabled by quantum computing in the near future, including faster, safer, and cheaper ways to design drugs and vaccines.

Quantum computing, through its diverse pathways, is also accessible to people beyond high school. So far, most university quantum computing programs are found at the graduate level, although individual courses in the area are offered in more and more undergraduate programs. In addition, larger companies in the area have their own training programs for new employees, as few highly qualified quantum computing personnel are available on the job market.

We anticipate that in the near future quantum computing will enter undergraduate education at a broad level, be it as stand-alone programs or options that can be chosen when pursuing a degree in science or engineering.

¹⁹https://k12cs.org/

A. TEACHING MATERIALS

All materials related to our workshops can be accessed on GitHub: https://github.com/pangara/TQE-HighTechQ.

ACKNOWLEDGMENT

The authors would like to thank Callysto Canada and High-TechU for supporting the development and delivery of our quantum computing workshops, and Honeywell Quantum Solutions for supporting Tom Markham. Copies of the board game *Entanglion* were donated by IBM Research Yorktown Heights. Quirklogic, Inc. provided access to Papyr and Quilla eInk devices, Quirklogic software as well as software support. Authors would also like to thank J. Knodel and C. Volin (Honeywell Quantum Solutions), M. Li (The Knowledge Society), P. Pashaei (University of British Columbia), as well as members from the Rigi-Pita Group at the University of Victoria and S. Joshi for their contributions to the quantum workshops. Our research has been approved by the University of Victoria Human Research Ethics Board under Protocol Number 20-0057.

REFERENCES

- M. N. Amin, R. P. Uhlig, P. P. Dey, B. Sinha, and J. Shatha, "The needs and challenges of workforce development in quantum computing," presented at 2019 Pacific Southwest Section Meeting, 2019. [Online]. Available: https://peer.asee.org/31846
- [2] B. Marr, "6 Practical examples of how quantum computing will change our world," 2017. [Online]. Available: https://bit.ly/3tmuZc1
- [3] D. Johnson, "Building a quantum computing workforce from the ground up," 2020. [Online]. Available: spectrum.ieee.org/nanoclast/at-work/ education/building-a-quantum-computing-workforce-from-the-ground-up
- [4] E. Grumbling and M. Horowitz, *Quantum Computing: Progress and Prospects*. Washington, DC, USA: National Academies of Sci., Eng., and Medicine, 2019. [Online]. Available: www.nap.edu/catalog/25196
- [5] S. Aaronson and D. Bacon, "Quantum computing and the ultimate limits of computation: The case for a national investment," 2008. [Online]. Available: bit.ly/2EV3LoG
- [6] M. G. Raymer and C. Monroe, "The US National Quantum Initiative," *Quantum Sci. Technol.*, vol. 4, no. 2, 2019, Art. no. 020504, doi: 10.1088/2058-9565/ab0441.
- [7] B. Sussman, P. Corkum, A. Blais, D. Cory, and A. Damascelli, "Quantum Canada," *Quantum Sci. Technol.*, vol. 4, no. 2, 2019, Art. no. 020503, doi: 10.1088/2058-9565/ab029d.
- [8] Z. Busch, "Germany invests \$717 million in IBM quantum computing," 2019. [Online]. Available: https://research.g2.com/insights/germany-invests-millions-ibm-quantum-
- [9] A.-F. Pelé, "French president details € 1.8b quantum plan." 2021. [Online]. Available: www.eetimes.eu/french-president-details-e1-8b-quantumplan/
- [10] R. P. Feynman, "Simulating physics with computers," Int. J. Theor. Phys, vol. 21, no. 6/7, pp. 467–488, 1982.
- [11] D. Deutsch, "Quantum theory as a universal physical theory," *Int. J. Theor. Phys.*, vol. 24, no. 1, pp. 1–41, 1985, doi: 10.1007/BF00670071.
- [12] D. Deutsch and R. Jozsa, "Rapid solution of problems by quantum computation," *Proc. Roy. Soc. London*, vol. 439, pp. 553–558, 1992, doi: 10.1098/rspa.1992.0167.
- [13] C. H. Bennett and S. J. Wiesner, "Communication via one- and two-particle operators on Einstein-Podolsky-Rosen states," *Phys. Rev. Lett.*, vol. 69, no. 2881, pp. 2881–2884, 1992, doi: 10.1103/Phys-RevLett.69.2881.
- [14] C. H. Bennett, G. Brassard, C. Crépeau, R. Jozsa, A. Peres, and W. K. Wootters, "Teleporting an unknown quantum state via dual classical and Einstein-Podolsky-Rosen channels," *Phys. Rev. Lett.*, vol. 70, 1993, Art. no. 1895, doi: 10.1103/PhysRevLett.70.1895.

- [15] D. R. Simon, "On the power of quantum computation," in *Proc. 35th Symp. Found. Comput. Sci.*, 1994, pp. 116–123, doi: 10.1109/SFCS.1994.365701.
- [16] E. Bernstein and U. Vazirani, "Quantum complexity theory," in Proc. 25th Ann. ACM Symp. Theory Comput., 1993, pp. 11–20, doi: 10.1145/167088.167097.
- [17] D. Coppersmith, "An approximate Fourier transform useful in quantum factoring," IBM Research Division, Armonk, NY, USA, Rep. 19642, 1994.
- [18] E. Bernstein and U. Vazirani, "Quantum complexity theory," *SIAM J. Comput.*, vol. 26, no. 5, pp. 1411–1473, 1997, doi: 10.1137/S0097539796300921.
- [19] P. W. Shor, "Algorithms for quantum computation: Discrete logarithms and factoring," in *Proc. 35th Ann. Symp. Found. Comput. Sci.*, 1994, pp. 124–134, doi: 10.1109/SFCS.1994.365700.
- [20] L. K. Grover, "A fast quantum mechanical algorithm for database search," in *Proc. 28th Ann. ACM Symp. Theory Comput.*, 1996, pp. 212–219, doi: 10.1145/237814.237866.
- [21] "Google quantum computing." 2021. [Online]. Available: research. google/research-areas/quantum-computing/
- [22] "Rigetti think quantum." 2021. [Online]. Available: www.rigetti.com
- [23] "Microsoft quantum." 2021. [Online]. Available: azure.microsoft.com/en-ca/ solutions/quantum-computing/
- [24] "IBM quantum." 2021. [Online]. Available: www.ibm.com/quantum-computing
- [25] "Xanadu quantum cloud." 2021. [Online]. Available: www.xanadu.ai/
- [26] "D-Wave." 2021. [Online]. Available: www.dwavesys.com/
- [27] "Honeywell quantum solutions." 2021. [Online]. Available: www.honeywell.
- com/us/en/company/quantum
- [28] "Qiskit." 2021. [Online]. Available: www.qiskit.org
- [29] "D-Wave Leap²." 2021. [Online]. Available: www.dwavesys.com/take-leap
- [30] Microsoft Azure, "Microsoft quantum development kit." 2021. [Online]. Available: www.microsoft.com/en-us/quantum/development-kit
- [31] "Google cirq." 2021. [Online]. Available: quantumai.google/cirq
- [32] "Rigetti forest SDK." 2021. [Online]. Available: qcs.rigetti.com/sdk-downloads
- [33] Xanadu, "Strawberry fields." 2021. [Online]. Available: strawberryfields.ai
- [34] J. Preskill, "Quantum computing in the NISQ era and beyond," *Quantum*, vol. 2, p. 79, 2018, doi: 10.22331/q-2018-08-06-79.
- "How build quantum [35] L. Converso. to G а comworkforce." puting 2020. [Online]. Available: www.accenture.com/us-en/blogs/accenture-research/ how-to-build-a-quantum-computing-workforce
- [36] A. Asfaw, "Quantum computing education must reach a diversity of students," 2020. [Online]. Available: www.ibm.com/blogs/research/2020/ 07/future-quantum-workforce/
- [37] A. Lamprou and A. Repenning, "Teaching how to teach computational thinking," in *Proc. 23rd Ann. ACM Conf. Innov. Techn. Comput. Sci. Educ.*, 2018, pp. 69–74, doi: 10.1145/3197091.3197120.
- https://research.g2.com/insights/germany-invests-millions-ibm-quantum-comptingJ. M. Wing, "Computational thinking," Commun. ACM, vol. 49, no. 3,A.-F. Pelé, "French president details € 1.8b quantum plan." 2021. [Online].pp. 33–35, 2006, doi: 10.1145/1118178.1118215.
 - [39] J. Sargent, "There's a diversity problem in the tech industry and it's not getting any better," 2019. [Online]. Available: https://bit.ly/3tl6YSO
 - [40] S. Hyder, "Why A diverse tech sector is the only healthy tech sector," 2020. [Online]. Available: www.forbes.com/sites/shamahyder/ 2020/10/27/why-a-diverse-tech-sector-is-the-only-healthy-tech-sector/? sh=3416c510747f
 - [41] V. Hunt, D. Layton, and S. Prince, "Why diversity matters," 2015. [Online]. Available: www.mckinsey.com/business-functions/ organization/our-insights/why-diversity-matters
 - [42] Microsoft, "The five years in every U.K. girl's life that decides if she will work in STEM," 2017. [Online]. Available: news.microsoft. com/en-gb/2017/03/01/40431/
 - [43] S. Aaronson, Quantum Computing Since Democritus. Cambridge, U.K.: Cambridge Univ. Press, 2013.
 - [44] W. Dür and S. Heusler, "Visualization of the invisible: The qubit as key to quantum physics," *Phys. Teacher*, vol. 52, no. 8, pp. 489–492, 2014, doi: 10.1119/1.4897588.

- [45] A. López-Incera and W. Dür, "Entangle me! A game to demonstrate the principles of quantum mechanics," *Amer. J. Phys.*, vol. 87, no. 2, pp. 95–101, 2019, doi: 10.1119/1.5086275.
- [46] H. R. Sadaghiani, "Spin first vs. position first instructional approaches to teaching introductory quantum mechanics," in *Proc. Phys. Educ. Res. Conf.*, 2016, pp. 292–295. [Online]. Available: https://meetings.aps.org/Meeting/APR17/Event/285989
- [47] A. MacLean and U. Stege, "HighTechU, Univ. of Victoria." 2019. [Online]. Available: hightechu.ca
- [48] T. Bell, I. Witten, and M. Fellows, "CS unplugged," 2015. [Online]. Available: https://classic.csunplugged.org/wp-content/uploads/2015/ 03/CSUnplugged_OS_2015_v3.1.pdf
- [49] "Q2Work program." 2021. [Online]. Available: q12education.org/
- [50] "Google quantum." 2021. [Online]. Available: https://quantumai.google/
- [51] R. S. Sutor, *Dancing With Qubits*. Birmingham, U.K: PACKT Publishing, 2019.
- [52] S. E. Economou, T. Rudolph, and E. Barnes, "Teaching quantum information science to high-school and early undergraduate students," 2020. [Online]. Available: arxiv.org/abs/2005.07874
- [53] B. Shoshany, ""Thinking quantum": Lectures on quantum theory," 2018. [Online]. Available: arxiv.org/abs/1803.07098
- [54] M. Mykhailova and K. M. Svore, "Teaching quantum computing through a practical software-driven approach experience report," in *Proc.* 51st ACM Tech. Symp. Comp. Sci. Edu., 2020, pp. 1019–1025, doi: 10.1145/3328778.3366952.
- [55] D. E. Knuth, "Literate programming," *Comput. J.*, vol. 27, no. 2, pp. 97–111, 1984, doi: 10.1093/comjnl/27.2.97.
- [56] A. Asfaw et al., "Learn quantum computation using Qiskit," IBM, 2020. [Online]. Available: community.qiskit.org/textbook
- [57] "IBM quantum experience." 2021. [Online]. Available: https:// quantum-computing.ibm.com/
- [58] A. Asfaw, "Coding with qiskit," 2019. [Online]. Available: bit.ly/33KYKJs
- [59] J.-S. Kim, "Coding with qiskit II," 2020. [Online]. Available: youtu.be/WvZvGTPdmjE
- [60] C. C. Tappert, R. I. Frank, I. Barabasi, A. M. Leider, D. Evans, and L. Westfall, "Experience teaching quantum computing," presented at the 52nd Ann. Meeting Assoc. Supporting Comput. Users Educ., 2019. [Online]. Available: https://eric.ed.gov/?id=ED597112
- [61] A. Perry, R. Sun, C. Hughes, J. Isaacson, and J. Turner, "Quantum computing as a high school module," FERMILAB, 2019. [Online]. Available: arxiv.org/abs/1905.00282
- [62] Y. Billig, Quantum Computing for High School Students. Bellevue, WA, USA: Amazon, 2018.
- [63] D. Sabol, A. Leider, and J. Glinka, "Quantum computing the easy way." 2021. [Online]. Available: bit.ly/3ipqCXX
- [64] S. Barabasi *et al.*, "Student user experience with the IBM Qiskit quantum computing interface," in *Proc. Adv. Inf. Commun.*, 2020, vol. 70, pp. 547–563, doi: 10.1007/978-3-030-12385-7_41.
- [65] O. Salehi, Z. Seskir, and I. Tepe, "A computer science-oriented approach to introduce quantum computing to a new audience," *IEEE Trans. Educ.*, 2021.
- [66] T. Bell, I. Witten, and M. Fellows, "Computer science unplugged: Off-line activities and games for all ages (original book)," 1998. [Online]. Available: bit.ly/3gzXAUS

- [67] T. Bell and J. Vahrenhold, "CS Unplugged—How Is It Used, and Does It Work? in Adventures Lower Bounds Higher Altitudes, vol. 11011, Cham, Switzerland: Springer, 2018, pp. 497–521.
- [68] T. Nishida, Y. Idosaka, Y. Hofuku, S. Kanemune, and Y. Kuno, "New methodology of information education with Computer Science Unplugged," in *Proc. 3rd Int. Conf. Informat. Secondary Sch.*— Evol. Perspectives, 2008, pp. 241–252, doi: 10.1007/978-3-540-69924-8_22.
- [69] R. Taub, M. Armoni, and M. Ben-Ari, "CS unplugged and middle-school students' views, attitudes, and intentions regarding CS," ACM Trans. Comput. Educ., vol. 12, no. 2, 2012, Art. no. 8, doi: 10.1145/2160547.2160551.
- [70] S. Carruthers, T. Milford, T. Pelton, and U. Stege, "Moving K-7 education into the information age," in *Proc. 15th Western Can. Conf. Comput. Educ.*, 2010, doi: 10.1145/1806512.1806529
- [71] S. Carruthers, "Grasping graphs," M.Sc. thesis, Department of Computer Science, Univ. of Victoria, Victoria, BC, Canada, 2010.
- [72] B. R. Rodriguez, "Assessing computational thinking in CS unplugged activities," M.Sc. thesis, Department of Computer Science, Colorado School of Mines, Golden, CO, USA, 2015.
- [73] B. Rodriguez, S. Kennicutt, C. Rader, and T. Camp, "Assessing computational thinking in CS unplugged activities," in *Proc. ACM Tech. Symp. Comput. Sci. Educ.*, 2017, pp. 501–506, doi: 10.1145/3017680.3017779.
- [74] B. Kim, H. Park, and Y. Baek, "Not just fun, but serious strategies: Using meta-cognitive strategies in game-based learning," *Comput. Educ.*, vol. 52, no. 4, pp. 800–810, 2009, doi: 10.1016/j.compedu.2008.12.004.
- [75] M. Romero, M. Usart, M. Ott, J. Earp, S; de Freitas, and S. Arnab, "Learning through playing for or against each other? Promoting collaborative learning in digital game based learning," in *Proc. 20th Eur. Conf. Inf. Syst.*, 2012. [Online]. Available: https://aisel.aisnet.org/ecis2012/93
- [76] J. Wootton, "Introducing the world's first game for a quantum computer," 2017. [Online]. Available: link.medium.com/QXFwsW85H6
- [77] J. Wootton, "Quantum battleships." 2017. [Online]. Available: bit.ly/3ijFC9v
- [78] J. D. Weisz, M. Ashoori, and Z. Ashktorab, "Entanglion: A board game for teaching the principles of quantum computing," in *Proc. Ann. Symp. Comput.- Hum. Interact. Play*, 2018, pp. 535–548, doi: 10.1145/3242671.3242696.
- [79] "Quantum games." 2016. [Online]. Available: bit.ly/3kqZ4TB
- [80] "Quantum cats." 2015. [Online]. Available: quantumcats.ca
- [81] "Decodoku." 2019. [Online]. Available: citizensciencegames.com/games/ decodoku/
- [82] "Teaching strategies and classroom complexity." 2021. [Online]. Available: bctf.ca/NewTeachers.aspx?id=31830
- [83] L. Westfall and A. Leider, "Teaching quantum computing," in Proc. Future Technol. Conf., 2018, pp. 63–80, doi: 10.1007/978-3-030-02683-7_6.
- [84] I. L. Chuang, N. Gershenfeld, and M. Kubinec, "Experimental implementation of fast quantum searching," *Phys. Rev. Lett.*, vol. 80, no. 15, 1998, Art. no. 3408, doi: 10.1103/PhysRevLett.80.3408.
- [85] K. Wehden, I. Faro, and J. Gambetta, "IBM's roadmap for building an open quantum software ecosystem," 2021. [Online]. Available: www.ibm.com/blogs/research/2021/02/quantum-development-roadmap/
- [86] D. Touretzky, C. Gardner-McCune, F. Martin, and D. Seehorn, "Envisioning AI for K-12: What should every child know about AI?," in *Proc. AAAI Conf. Artif. Intell.*, 2019, vol. 33, no. 1, pp. 9795–9799, doi: 10.1609/aaai.v33i01.33019795.